Page 1

This schedule is subject to change a. Lowest homework (HW) score will be dropped. Each HW contains 2-5 problems. b. Group project: Intermediate due dates will be established. The projects will be demo’ed at the end of the course.

CS 425 Database Systems Spring 1996 Semester Schedule

Week Day Topic Readings Events

1 TU TH

Introduction
E-R Model

1-2 3 2 TU TH

E-R Model
Relational Model

3, 21.1 6-6.4

HW1^a

3 TU TH

Relational Model/SQL
Relational Calc/QUEL/QBE

                            6.5-6.7,7-7.4

8-8.4

HW2 (HW1 Due)

4 TU TH

Ntwk & Hierarchical Models
Database Design

10-10.3,11-11.3 12 HW3 (HW2 Due)

5 TU TH

Database Design
Database Design

12 12, 13-13.2

HW4 (HW3 Due)

6 TU TH Review Mid-Term

(HW4 Due)

Spring Break

7 TU TH

File Organization
Indexing

4-4.4, 4.8 5-5.3 Project Assigned^b

8 TU TH

Discussion on Project
Query Processing/Optimiz. 16-16.3

Project Grps Due HW5 9 TU TH

Transaction Processing
Transaction Processing

17 17 HW6 (HW5 Due)

10 TU TH

Concurrency Control
Recovery

18 19 HW7 (HW6 Due)

11 TU TH

Object Oriented Databases
Distributed Databases (HW7 Due)

12 TU TH

Case Study (Teradata/BeSS)

Review

Three Week Reading Time Demo Projects Here

Final Exam Cumulative